Author’s response to reviews

Title: Identification and Ranking of Recurrent Neo-Epitopes in Cancer

Authors:

Eric Blanc (eric.blanc@bihealth.de)
Manuel Holtgrewe (manuel.holtgrewe@bihealth.de)
Arunraj Dhamodaran (arunraj.dhamodaran@mdc-berlin.de)
Clemens Messerschmidt (clemens.messerschmidt@bihealth.de)
Gerald Willimsky (gerald.willimsky@chaquite.de)
Thomas Blankenstein (tblanke@mdc-berlin.de)
Dieter Beule (dieter.beule@bihealth.de)

Version: 2 Date: 23 Sep 2019

Author’s response to reviews:

We thank reviewer 2 for her/his comments, which help us improve the manuscript. Following her/his suggestions, we have:

- Added a sentence to the "Data set" subsection of the Methods, to highlight the overall workflow process,

- Changed and extended the "Epitope selection & MHC Class I binding prediction" subsection to clarify our usage of NetMHCcons, and have added a reference which supports it.

We believe that the Methods section now explicitly details all the steps taken for our analysis.